Author’s response to reviews

Title: Cytomegalovirus immune reconstitution inflammatory syndrome manifesting as acute appendicitis in an HIV-infected patient

Authors:

Kimberly F Faldetta (kimberly.faldetta@nih.gov)
Sarah Kattakuzhy (sarah.kattakuzhy@nih.gov)
Hao-Wei Wang (hao-wei.wang@nih.gov)
Irini Sereti (isereti@niaid.nih.gov)
Virginia Sheikh (sheikhv@niaid.nih.gov)

Version: 2  Date: 28 February 2014

Author’s response to reviews:

In response to the BMC Medicine editor’s suggestion that our manuscript would be best suited for the scope and readership BMC Infectious Diseases, we have reformatted our manuscript to meet the specifications and requirements of submission to this journal.